Author's response to reviews

Title: The burden of co-existing dermatological disorders and their tendency of being overlooked among patients admitted to Muhimbili National hospital in Dar es Salaam, Tanzania

Authors:

Yassin M Mgonda (ymgonda@muhas.ac.tz)
Pauline N Chale (kossapn@yahoo.com)

Version: 4 Date: 8 April 2011

Author's response to reviews:

All revisions as suggested by the editorial team have been made.

Major revisions:
Author list: The list of authors in the manuscript has been written exactly as it is in the submission system

Tables: all tables have been created using a Tables tool and are not linked files. I have ensured that the text alignment in columns is clear and consistent. All vertical lines in the tables have been removed

Minor revisions:
All track changes have been removed. Typographical and other necessary final revisions have been made